Author's response to reviews

Title: Acute left main coronary artery thrombosis due to cocaine use

Authors:

EFSTRATIOS APOSTOLAKIS (stratisapostolakis@yahoo.com)
GRIGORIOS TSIGKAS (gregtsig@hotmail.com)
NIKOLAOS G BAIKOUSSIS (ngbaik@yahoo.com)
IOANNA KONIARI (iokoniari@yahoo.gr)
DIMITRIOS ALEXOPOULOS (baikoussisn@yahoo.com)

Version: 3 Date: 6 August 2010

Author's response to reviews:

Cover Letter-Patras 2010-08-06
Dear Editor,

We would send you our final corrections of our accepted article entitled "Acute left main coronary artery thrombosis due to cocaine use" by Apostolakis et al. Competing interests' section, patient's consent and figure legend are added in the manuscript.

All the best
Corresponding author:
Nikolaos G. Baikoussis MD
Cardiothoracic Surgeon
Patras University Hospital
Rion Patras 26500 Greece
Tel: +306974201838
e-mail: ngbaik@yahoo.com